Author’s response to reviews

Title: Non-Hispanic Whites have higher risk for pulmonary impairment from pulmonary tuberculosis.

Authors:

Jotam G Pasipanodya (jotam.pasipanodya@UTSouthwestern.edu)
Edgar Vecino (Edgar.Vecino@unthsc.edu)
Thaddeus L Miller (thaddeus.miller@unthsc.edu)
Guadalupe Munguia (Guadalupe.Munguia@unthsc.edu)
Gerry Drewyer (Gerry.Drewyer@tarrantcounty.com)
Michel Fernandez (Michel.Fernandez@unthsc.edu)
Philip Slocum (pslocum@atsu.edu)
Stephen E Weis (weistephen@me.com)

Version: 6 Date: 19 January 2012

Author’s response to reviews: see over
To the Editor through Mr. Victorino Silvestre,

RE: MS: 2869734425528295

Research article: Non-Hispanic Whites have higher risk for pulmonary impairment from pulmonary tuberculosis.

We have made the requests made by your office regarding the layout of the title page and have completed of the ‘Competing interests’ section.

The current email addresses for all authors are also included.

Sincerely,

Stephen E. Weis., D.O.